Author's response to reviews

Title: The Relationship between Fasting Plasma Glucose and MPO in Patients with Acute Coronary Syndrome

Authors:

Xiangyu Zhang (xiangyuzhang2001@yahoo.com.cn)
Lini Dong (donglini@163.com)
Qiong Wang (wangqiong25339@sina.com)
Xiaomei xie (marrie2088@sina.com)

Version: 2 Date: 28 November 2014

Author's response to reviews:

Dear Sr.
Thank you for your comments. I have already made changes. Point to point changes as blow#

1#Title page has already contaied the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
2#1*Corresponding author: Xiaomei Xie is only author should be linked.
3#I have already revised my manuscript include line and page numbers
4#I have placed Figure Legends after the references list and tables after the Figure Legends.
5#I have arranged the sections of Abbreviations#Competing Interests#Authors’ Contributions and Acknowledgements as follows after the "Conclusions" section.

Best regards
xiaomei xie
5)